36-401 Homework #11 Due 11/30/00

1. (15 pts) RwG problems 5, RwG p. 213 (data in kob.dat).

2. (10/100 Project points) One member of your project group must send me an e-mail listing the major (in terms of your time) aspects of your progress on the project for the week. Send this by 2 PM Thursday. A different group member must do this again on 12/7. In addition your group must meet with me between 11/27 and 12/6. Your group project write-up (20% of your grade; 80/100 Project points) is due 3 PM Thursday 12/12. If necessary, individuals may write a brief “dissenting opinion”; otherwise all group members receive the same grade for the group project.